Author's response to reviews

Title: A study of home deaths in Japan from 1951 to 2002

Authors:

Limin Yang (yanglm@med.juntendo.ac.jp)
Naoko Sakamoto (sakana@med.juntendo.ac.jp)
Eiji Marui (marui@med.juntendo.ac.jp)

Version: 7 Date: 2 March 2006

Author's response to reviews:

Cover letter

Re: Article No. 1276320553849831 - A study of home deaths in Japan from 1951 to 2002

For the third revised manuscript.

Dear Sir:
Thank you for your kind letter of FEB 24, 2006. We are happy to know that our paper entitled "A study of home deaths in Japan from 1951 to 2002" (Article No. 1276320553849831) will be acceptable. We have revised the title as 'A study of home deaths in Japan from 1951 to 2002'. The competing interests have been written as 'The authors declare that they have no competing interests' and author's contributions have been improved as request. We hope that these revisions are satisfactory and that the revised version will be acceptable for publication in BMC palliative care.

Sincerely yours
Limin Yang
Jan 22, 2006